Omega 🛛 (📲📲) 📲🗕



Contents lists available at ScienceDirect

Omega



journal homepage: www.elsevier.com/locate/omega

Scheduling models for optimal aircraft traffic control at busy airports: Tardiness, priorities, equity and violations considerations $\stackrel{\mathackar}{\sim}$

Marcella Samà, Andrea D'Ariano*, Paolo D'Ariano, Dario Pacciarelli

Università degli Studi Roma Tre, Dipartimento di Ingegneria, via della Vasca Navale, 79, 00146 Rome, Italy

ARTICLE INFO

Article history: Received 5 March 2015 Accepted 10 April 2016

Keywords: Aviation Air traffic optimization Efficient landing and take-off operations Microscopic scheduling models Mixed integer linear programming Job shop scheduling

ABSTRACT

This work addresses the real-time optimization of take-off and landing operations at a busy terminal control area in case of traffic congestion. Terminal areas are becoming the bottleneck of the entire air traffic control system, in particular in the major European airports, where there is a limited possibility to build new infrastructure. The real-time problem of effectively managing aircraft operations is particularly challenging, since it is necessary to incorporate the safety regulations into the optimization model and to consider numerous performance indicators that are important to compute good quality solutions. However, in practice there is no well-recognized objective function and traffic controllers often use simple scheduling rules. In this paper, mixed integer linear programming formulations are proposed to investigate the trade-off between various performance indicators of practical interest, while taking into account the safety constraints with a high modeling precision. Experiments are performed for the two major Italian airports, Milano Malpensa and Roma Fiumicino, by simulating various sets of random landing and take-off aircraft disturbances. Practical-size instances are solved to (near)optimality via a commercial solver. The optimized solutions are also compared with a commonly used scheduling rule. A comprehensive computational analysis makes possible the selection of those solutions that are able to find a good compromise among the various indicators and, consequently, the investigation of the most representative formulation.

© 2016 Elsevier Ltd. All rights reserved.

1. Introduction

The ever growing demand of air transport is increasing the pressure on air traffic controllers, since air traffic in peak hours is getting closer to the capacity of the Terminal Control Area (TCA), at least in the major European airports where there is limited possibility of creating new infrastructure. Aviation authorities are thus seeking optimization methods to better use the available infrastructure [4,5,21,27,45]. However, the development and the implementation of effective optimization methods for such operational problems require the consideration of a number of aspects that are rarely taken into account simultaneously in the related scheduling theory:

• The optimization model should be able to incorporate all detailed information that is compliant with the safety regulations of the TCA, including information which is not relevant for the air traffic flow management in large networks with multiple

 $\ensuremath{\,^{\ensuremath{\scriptscriptstyle \pm}}}$ This manuscript was processed by Associate Editor B. Kis.

* Corresponding author.

E-mail address: dariano@ing.uniroma3.it (A. D'Ariano).

http://dx.doi.org/10.1016/j.omega.2016.04.003 0305-0483/© 2016 Elsevier Ltd. All rights reserved. airports and is therefore neglected in macroscopic models [17,22,48]. In most of the macroscopic models, the characteristics of the airport infrastructure are drastically simplified and the flight paths are aggregated, so that *potential conflicts* between single aircraft may not be visible, at least at the level of runways, ground and air segments of the TCA. A potential conflict occurs whenever aircraft traversing the same resource do not respect the minimum required safety distance.

- The time available for developing a new schedule of take-off and landing aircraft in the TCA can be very limited, since a computerized scheduler should be able to promptly react to any significant change occurring during operations.
- To a large extent, air traffic control operations and related issues are still scheduled by human controllers, who develop feasible aircraft schedules in the TCA based on their past experience, intuition and some scheduling rules without using any formally defined performance indicator. Recently, the push from SESAR and for CDM compliance [36] is making this less common though and airports have at least some automated support systems for some of the operations. For example, different commercial arrival manager systems are used at various airports [42,74]. However, the controllers usually have to fine tune the arrivals sequencing coming out of the systems themselves at

M. Samà et al. / Omega **■** (**■■■**) **■■**–**■**■

the moment, since these systems do not usually (fully) take into account the fine details of the aircraft movement required to land in the correct order. Furthermore, we believe that further automated support is required in order to compute alternative (near)optimal ASP solutions and evaluate them in terms of a number of performance indicators in a short-term. In fact, the existing arrival manager systems incorporate various performance indicators that need to be fine tuned across all airports. The lack of a generally recognized performance indicator to optimize places importance on the definition of acceptable objective functions. The quality of scheduling aircraft in the TCA typically involves several performance indices reflecting the interests of the different actors involved in air traffic management, such as the aircraft punctuality, the utilization level of airport resources, the costs incurred by different airline companies in terms of delays, broken flight connections and energy consumption, and so on. All these indices should be taken into account in the schedule development phase.

This paper addresses the first item by developing mixed integer linear programming (MILP) formulations, that take into account the relevant TCA safety aspects and various performance indicators with a high level of detail. As shown in the survey of Bennell et al. [18], the aircraft scheduling literature presents numerous models of the independent runway sequencing problem. This problem is modelled as a single machine scheduling problem. A natural way to model and solve a more accurate and extended aircraft scheduling problem with interdependent runways and air segments of the TCA is via job shop scheduling. The latter type of modelling approach permits to consider the airspace interactions between aircraft in order to compute better quality aircraft scheduling solutions in terms of delay management and traffic flow coordination in the TCA.

The MILP formulations proposed in this paper can be considered as a generalization of existing job shop scheduling models with blocking (no-store), no-wait and other additional constraints. These models are known under the name *generalized disjunctive graph* or *alternative graph*. Previous research on those job shop scheduling models has been successfully applied to model and solve complex benchmark instances on job shop scheduling [39,55–57,62], railway traffic management problems [25,50,52], and air traffic management problems [20,27,28,65,66].

The second item suggests that optimization models with a single objective function are more suitable than multi-objective approaches, since more efficient tools are available to solve these problems. This is also the most common choice in the literature (see, e.g., the reviews in [14,18,24,46,48,61]).

The present paper investigates MILP formulations with single objective functions in order to find a good compromise among the different indices listed in the third item. Specifically, we observe that aircraft typically flies at constant speed in the TCA and that at constant speed the energy consumption is almost proportional to the flying time. We use the aircraft flow time as a surrogate of the energy consumption. Also, we adopt makespan-like objective functions in order to minimize the maximum completion time (i.e. the arrival time of the last aircraft), as a common surrogate for the throughput maximization, or the maximum tardiness (i.e. the largest aircraft delay). Moreover, we implicitly take into account the minimization of broken flight connections by minimizing the number of aircraft delayed more than a given threshold. All performance indicators can be measured in terms of aircraft arrival times at the entrance of the TCA and at the runways.

The aircraft scheduling problem (ASP), we deal with in this paper, can be summarized as follows. Given a set of landing/takeoff aircraft and for each aircraft its path in the TCA, its current position, its scheduled runway occupancy time and the required time window to accomplish the arriving/departing procedures, the ASP is to assign the start time to each aircraft in all the resources it crosses in its path in such a way that all the potential conflict situations between aircraft are solved (at a microscopic level) and a suitable objective function is minimized.

This work follows the approach of Bianco et al. [20], based on the no-wait version of the job shop scheduling problem. However, this paper is based on the alternative graph model introduced by Mascis and Pacciarelli [55], that is able to model the ASP with an increased level of detail. The higher modeling precision includes further relevant TCA aspects such as holding circles, waiting in flight before landing, traveling in feasible time windows, hosting multiple aircraft simultaneously in air segments and individual aircraft simultaneously in runways. Previous works based on the alternative graph model of the ASP have been proposed recently [26-28,65,66]. D'Ariano et al. [26,28] deal with the development of a branch and bound algorithm for the ASP. D'Ariano et al. [27] extend the ASP to a routing and scheduling problem and solve it with a tabu search algorithm. Samà et al. [65,66] develop a rolling horizon approach for the original and extended ASP. However, all these works deal with the minimization of a makespan-like objective function.

The contribution of this work is to generalize the work done on the ASP modelled via alternative graphs. We investigate microscopic MILP formulations of the ASP with different objective functions and examine the differences between the ASP solutions in terms of various performance indicators. As far as we know, the proposed formulations increase the level of detail regarding the modeling of the constraints in the airspace nearby the TCA compared to the existing models, and permit to deal with any kind of objective function and constraint. We believe that the investigation of a suitable formulation of the ASP, that takes into account several performance indicators and models the constraints with high precision, is still an open problem in the related literature.

A computational study is presented for assessing the practical applicability of the proposed formulations. The ASP solutions are analyzed from the viewpoint of the above described performance indicators and trade-off between them, while previous research often focuses on a single performance indicator, with a myopic view in terms of other possible performance indicators. A procedure is proposed to develop a *combined formulation* with a good trade-off performance on several indicators.

The experiments have been carried out on the main Italian airports in terms of passenger flows: Roma Fiumicino (FCO) and Milano Malpensa (MXP). Regarding the air traffic disturbances, 40 randomly delayed scenarios are considered for practical-size instances. The resulting problems are solved with a commercial solver to (near)optimality for each ASP formulation. The optimized solutions are also compared with the solutions computed by a practical scheduling rule.

Section 2 reviews the literature most relevant for this work. Section 3 formally describes the modelling of specific ASP constraints. Section 4 presents the mathematical formulations. Section 5 reports the experiments conducted on the FCO and MXP instances. Section 6 summarizes the paper results and outlines future research directions. Two appendices illustrate the alternative graph modeling and solving for a numerical ASP example.

2. Literature review

This section briefly reviews recent papers on some aspects of the air traffic flow management (ATFM) problem. We present various ATFM literature classifications and discuss our contribution. A more general discussion of the existing literature can be found e.g. in [1,14,18,24,33,46,48,61].

A first classification of the ATFM literature is based on the following two basic categories: the traffic control between airports (see e.g. [17,21,22]) and the traffic control in the TCA of an individual airport (see e.g. [20,26,37,45,51]). For the former category, macroscopic models for large networks with multiple airports and aggregated flight paths are often adopted. For the latter category, microscopic models are proposed with identification and resolution of potential aircraft conflicts at the level of ground and air resources. This paper deals with the development of microscopic models for the resolution of potential conflicts in the TCA.

A second classification is based on the type of information. When dealing with static information (see e.g. [26,28,34,63]), the position and the speed of all aircraft are known in the traffic prediction. The case with dynamic information (see e.g. [15,43,65,75]) requires the computation of an aircraft schedule every time a new incoming aircraft is known. This paper studies an ASP based on static information, since all the data are known before the optimizer starts the computation of the ASP solution. However, this approach can be inserted in a dynamic system that iteratively solves an ASP problem with static information (see e.g. [65,66]).

A third classification is based on the algorithmic approaches. Among the heuristic approaches, fast heuristics are proposed in [27,28,34,60]. Exact procedures can be found in [26,28,34,35]. This paper does not focus on the investigation of new ASP algorithmic approaches or mathematical properties. We are interested in the comparison of (near)optimal ASP solutions in terms of several performance indicators. In order to compute a (near)optimal solution for each performance indicator, we use a specific MILP formulation and a general MILP solver.

We now present a detailed review of some literature most related to our work, with specific reference to the different choice of the objective functions and constraints in the formulations. We organize the discussion based on the type of problem studied.

Allahverdi et al. [3] and Ball et al. [11] present a detailed panoramic of different solving approaches and objective functions used in the aircraft scheduling literature. In particular, the main objective functions discussed are the minimization of delays and costs. In fact, the costs are often calculated based on the deviation from the nominal schedule, i.e., in terms of aircraft delays.

In the arrival scheduling literature, Hu et al. [43] minimize the sum of the difference between the predicted and the allocated landing times of each aircraft. Eun et al. [35] try to limit the aircraft delays and the deviation from the estimated arrival time, by taking into account airline preferences. Ernst et al. [34] measure the cost associated with the deviation from the preferred aircraft landing time. Beasley et al. [15,16] study an aircraft displacement problem and consider the cost of a solution adjustment procedure where, if an aircraft is further delayed with respect to an initial solution, an additional penalty has to be paid. Sölveling et al. [73], instead, include in the cost function the environmental impact, in terms of fuel and CO_2 emissions, when there are deviations from the nominal schedule. Soomer and Franx [72] combine cost functions declared by the airlines for each aircraft typology and rescale them, making the resulting solution fair.

Artiouchine et al. [6] assign landing times to aircraft for a single runway by minimizing the use of holding patterns or by maximizing the minimum time elapsed between two consecutive landings. Hu and Di Paolo [44] also consider two objective functions: the minimization of the total airborne delay of all aircraft and the maximum length of all arrival queues. The first objective function emphasizes the operating cost of airlines, while the second objective function focuses more on the efficiency of using airport capacity. Sabar and Kendall [64] minimize the total landing deviations from a desired landing time. Other works on the landing and runway scheduling problems aim to maximize the use of airport capacity (i.e., the throughput). Bianco et al. [19,20] adopt the minimization of the maximum completion time for the throughput maximization, as previously described in Psaraftis [63]. Balakrishnan and Chandran [10] also adopted this objective function and discussed its practical importance in order to manage runway sequencing problems at major U. S.A. airports. Furthermore, they compare it with the minimization of the maximum aircraft delay and of the sum of aircraft delays.

In the departure scheduling literature, Marin [54] models the taxi planning problem as a multicommodity flow network problem with link capacities and minimizes total routing time for all the flights. Atkin et al. [7] focus on the development of an automated advisory system to help the runway controller to increase the throughput of the departure runway and to reduce the aircraft delay without negatively affecting safety and other feasibility constraints on the aircraft reordering. A weighted sum of delay, penalty and cost factors is minimized. The various factors in the objective function have an ordered degree of importance. Atkin et al. [8] present an improved automated advisory system in which further aspects of the take-off runway scheduling problem are taken into account, including a consideration of taxiway capacity congestion and some measure of the equity of delay. Atkin et al. [9] propose another objective function based on the minimization of a weighted sum of aircraft delays, equity of delay (division of the delay between aircraft), and take-off time re-negotiations. Clare and Richards [23] minimize a weighted combination of the total taxi time, the largest taxi time and the total taxi distance for a combined taxi and runway scheduling problem.

Another stream of research focuses on the development of a ground delay program. Kotynek and Richetta [47] define the main objective of a ground delay program as delaying aircraft before they depart from their origin airports, in case of a reduced landing capacity at the airport of destination. Several authors address the minimization of a weighted combination of ground delays and equity measures [12,13,38,47,58]. A common equity consideration is to generate tactical aircraft schedules via a first scheduled, first served rule, since the inequity is often measured as a deviation from an ideal (equitable) schedule. A more complex definition of the goal of a ground delay program is proposed in other works [49,53], including considerations of airline fuel costs, air traffic controller workloads, and passenger comfort, safety and equity.

We next discuss further multi-objective approaches for coordinating ATFM decisions. Sherali et al. [69] introduce an MILP formulation to generate flight-plans that satisfy various equity, workload, and safety considerations under different airspace scenarios. The considerations of the performance indicators are implemented via the addition of specific constraints and objective function components. In particular, they define equity measures that minimize the spread as well as the maximum measure of effectiveness over the airline carriers. Sherali et al. [70] present a definition of equity that takes better account of the distribution of individual airline schedule costs. Sherali et al. [71] perform sensitivity analysis of various scenarios and model parameters, including a study of the different weights of the objective function. Sherali et al. [68] minimize another weighted objective function with consideration of total system fuel, delay, and cancellation costs. Different equity considerations are introduced based on fuel and delay costs. Grushka-Cockayne et al. [40] present a multistakeholder, multicriteria decision-making framework for Eurocontrol in order to maximize an overall performance score, weighted by the stakeholders' importance weights. Andreatta et al. [5] study the trade-off between airborne holding delay and ground delay. Their objective function is the minimization of a weighted sum of the number of flights with an airborne holding or ground delay. Aktürk et al. [2] propose a conic quadratic

optimization approach to solve an aircraft recovery problem with minimization of the fuel consumption and the aircraft delay.

Most of the literature is based on a simplified model of the TCA. We believe that this paper is a step forward in the definition of a flexible microscopic formulation for the ASP in the TCA, that is able to take into account different performance indicators, either in the objective function or in the set of constraints. Our work is complementary to the work done in the departure/take-off sequencing literature. We use a detailed modeling of inter-aircraft separation rules for arrival and departure aircraft. However, our work does not deal with the detailed management of ground movements. Our contribution is the coordinated management of the runways and the air segments of TCA. Specifically, our work is not the classical departure and take-off sequencing problem that usually schedules aircraft in independent runways, and thus can be viewed as a set of single machine scheduling problems. We deal with a problem with interdependent runways and air segments of the TCA. As reported in the existing literature (see e.g. the survey paper [18]), the problem studied in this paper can be viewed as a job shop scheduling problem, that is by far more complex than the independent runway scheduling problem.

We think that the traffic controllers should be informed of the existence of alternative ASP solutions in real-time and of the potential impact generated by implementing each ASP solution in terms of a number of relevant performance indicators. To this end, this paper studies some of the above discussed indices and their combinations. We evaluate them with special emphasis on extent of the impact of each specific indicator on the others. The evaluation is performed on MILP formulations based on the alternative graph model of [55], originally developed for a makespanlike objective function. In fact, previous works dealing with the alternative graph model of the ASP (see e.g. [26-28,65,66]) are focused on the minimization of the maximum consecutive delay in a busy TCA (i.e. the minimization of the largest delay caused by the resolution of potential conflicts between aircraft traveling in a busy TCA during a given time horizon of traffic optimization). In this work, we generalize the alternative graph model of the ASP, in order to deal with any kind of performance indicator. Practical-size instances are solved by an MILP solver to (near)optimality.

3. Problem description

In the TCA, landing (arriving) aircraft move from an air entry point of the TCA to a runway via landing air segments, following a standard descent profile, while maintaining a minimum safety distance between every pair of consecutive aircraft, depending on their type and position (at the same or different altitude). Final approach spacing tools can be of support for the computation of feasible sequencing, moving and spacing of landing aircraft in the TCA [31].

Similarly, take-off (departing) aircraft leave the runway flying toward the assigned exit point via take-off air segments along a standard ascent profile, still respecting the minimum safety distance (safety separation). The space distance can be translated into a time distance, *setup time*, by taking into account the different aircraft speeds. Setup times are considered sequence-dependent, since the minimum distance between different aircraft categories (heavy, medium, light and others) depends on the relative processing order of the common resources. For instance, the distance between heavy and light aircraft is much larger when light aircraft follows heavy aircraft than vice versa. Setup times do not only depend on the aircraft times but also on the route chosen for each aircraft.

Each aircraft has an assigned entry time into the TCA, which is the minimum time, *release time*, the landing/take-off procedure can start according to the current aircraft position and speed. Each aircraft has scheduled times, *due date times*, to start processing some TCA resources. Eventually, aircraft can also have a maximum time, *deadline time*, to start processing some TCA resources.

The runway is a blocking (no-store) resource [41] since it can only be occupied by one aircraft at a time, while the air segment is an uncapacitated resource since several aircraft can occupy it at the same time provided that the required safety separations are satisfied. Each aircraft has a *processing time* on each runway sand on the air segments before or after it, according to its landing/ take-off profile. On the air segments, the processing time varies between a pre-defined time window, due to a limited possibility of aircraft speed changes.

Once an arriving aircraft enters the TCA, it should proceed to the runway. However, before entering the airport area, *airborne holding time* can be used to make aircraft waiting in flight until they can be guided through their landing procedure, that means flying in circles in specific areas named *holding circles*. On entry to a holding circle, the aircraft must fly at a fixed speed for a number of half circles, as prescribed by the air traffic controller. We assume that there are no aircraft sequencing decisions in the holding circle. Therefore, the aircraft can exit in a different order they enter in the holding circle and each holding circle resource is uncapacitated.

Departing aircraft can be delayed in entering the TCA at ground level, i.e. before entering the runway. A departing aircraft is supposed to take-off within its assigned time window and is late whenever it is not able to accomplish the departing procedure within its assigned time window. Following the procedure commonly adopted by air traffic controllers, we consider a time window for take-off between 5 min before and 10 min after the *Scheduled Take-off Time* (STT). A departing aircraft is considered delayed in exiting the TCA if leaving the runway after 10 min from its STT. We assume that all take-off aircraft have a take-off time window with the same tolerance. Arriving aircraft are late if landing after their *Scheduled Landing Time* (SLT).

We use the following notation for the aircraft delays. *Entrance delay* is the delay of a landing/take-off aircraft on entry to the TCA. *Total exit delay* is the delay of a landing/take-off aircraft at the entrance in the runway/take-off air segment resource. The latter value is partly a consequence of a possible late entrance, which causes an *unavoidable delay* at the runway/take-off air segment, and partly due to additional delays caused by the resolution of potential aircraft conflicts in the TCA, which is named *consecutive delay* [25–27].

A landing aircraft can have a consecutive delay at the entrance resource, if it is delayed in entering the TCA due to other aircraft scheduled on its entrance landing air segment. Landing and takeoff aircraft can have a consecutive delay on a runway, if they have to give precedence to other aircraft in one or more TCA resources. These consecutive delays will be used to formulate the objective functions considered in this work.

3.1. Performance indicators

The typical objectives of any real-time scheduling practitioner are (i) to find a good schedule in the short-term while (ii) trying to avoid negative long-term effects of the rescheduling decisions. In the air traffic environment under study, these needs correspond to schedule arrivals and departures with the aim of (i) reducing short-term delays and (ii) trying to recover the off-line plan of arrivals-departures as quickly as possible, in order to reduce longterm propagation of perturbations.

In this paper, the minimization of the maximum tardiness takes into account the first goal, while the minimization of the maximum completion time is a surrogate for the second goal. In

M. Samà et al. / Omega **■** (**■■■**) **■■**–**■**■



Fig. 1. Malpensa (MXP) terminal control area.

general, the minimization of a makespan-based objective function results in a more compact schedule compared to other practical objectives, which leaves more time available for accommodating future arrivals and departures and tries to reduce the long-term effects of past perturbations.

In real-time the ASP translates into an optimization problem for a limited time horizon of traffic optimization, with consequent myopic view on the overall traffic control horizon. In a companion paper [65], we deal with a rolling horizon approach in order to investigate the impact of traffic optimization in terms of the overall traffic control horizon. However, as far as we know, the rolling horizon approaches, or other problem decomposition approaches, cannot compute the optimal ASP solution for largescale problems and thus are not useful in the context of this work.

There can be many ASP solutions with different view points (airline companies, local and global authorities). We choose a set of objective functions to be minimized in a given time horizon of traffic optimization, each one interested in looking at a particular aspect of the ASP.

Our interpretation of equity is different from the common equity consideration in the related literature that is based on some scheduling rule, such as the first scheduled, first served rule or the first come, first served rule. We agree that these rules can be considered useful measures of the deviation from an ideal (equitable) schedule. However, we observe that using such a scheduling rule may generate poor quality solutions in terms of the minimization of delay propagation or even infeasible schedules when applied to reschedule aircraft during operations. For these reasons, we use a new equity consideration that is viewed as the computation of an ASP solution that minimizes the consecutive delays, i.e. the delays due to the aircraft sequencing decisions. Specifically, we consider two equity measures: the minimization of the maximum consecutive delay (maximum tardiness), and the minimization of the largest difference of consecutive delays among the aircraft of each class (priority equity).

When minimizing the maximum tardiness, we focus on the minimization of the longest path in the graph, that often involves a few aircraft. The minimization of the average consecutive delays (average tardiness) is a more global vision since it requires the consideration of all delayed aircraft.

In our approach, *aircraft priorities* are also taken into account as follows: landing aircraft, due to safety measures, have greater priority than departing aircraft, which can wait at ground level with fewer risks. A landing, delayed aircraft has even greater priority, due to a lower level of fuel. In fact, a *fairness* concept in air traffic management is to give priority to late aircraft [14]. In this paper, the set of aircraft is divided into four classes which are ordered below in importance (from the highest to the lowest priority): (1) landing, delayed aircraft; (2) landing aircraft on time; (3) take-off, delayed aircraft; (4) take-off aircraft on time.

minimization of the weighted average consecutive delays (*priority tardiness*) requires a weighted consideration of the aircraft.

Mirroring the two different approaches used to minimize delays, in this priority scenario we compare the solutions that take into account equity, defined here as the minimization of the average difference between maximum and minimum tardiness for each class, with the ASP solutions that minimize a weighted average tardiness with weights assigned to each aircraft according to the corresponding level of priority.

Another aspect to be taken into account when solving the ASP is the use of existing, critical airport resources. This consideration translates into the maximization of the throughput, that can be viewed as the minimization of the maximum completion time (see e.g. [20]). In our view, it corresponds to the landing/take-off time of the last aircraft traveling in the TCA during the time horizon (time span) of the traffic optimization. This objective function is compared with the minimization of the average completion time. The latter objective looks at all aircraft in the studied time horizon of traffic optimization.

Finally, we evaluate the number of delayed aircraft exceeding a given tolerance thresholds and minimize the number of *deadline violations*, that imply additional operational costs for the airline companies, due e.g. to broken flight connections with other aircraft at the same TCA. This objective is focused on the aircraft that have a delay above the given threshold (*tardy jobs*).

3.2. Terminal control areas

Fig. 1 shows the TCA scheme of Milano Malpensa airport (MXP). There are two interdependent runways (RWY 35L, RWY 35R), used both for departing and arriving procedures. The MXP resources are three airborne holding circles (resources 1–3 in Fig. 1, named TOR, MBR, SRN), 11 air segments for arriving procedures (resources 4–14), a common glide path (resource 15), two runways (resources 16 and 17) and three air segments for departing procedures (resources 18–20, named SRN, TELVA, RMG). The common glide path resource includes two parallel air segments before the runways for which traffic regulations impose a minimum diagonal distance between landing aircraft added to a minimum longitudinal one.

Fig. 2 presents the scheme of another TCA, Roma Fiumicino airport (FCO). In this case, three interdependent runways (RWY 16L, RWY 16R, RWY25) can be used for departing and arriving procedures, but two of them (RWY 16R and RWY 25) cannot be used simultaneously and are thus considered as one. The FCO resources are three airborne holding circles (resources 1–3 in Fig. 2, named CIA, CMP, TAQ), seven landing air segments (resources 4–10), two runways (resources 12 and 13), a common glide path (resource 11) and three take-off air segments (resources 14–16, named BOL, RAVAL, ELIVIN).

M. Samà et al. / Omega **■** (**■■■**) **■■**–**■**■



Fig. 2. Fiumicino (FCO) terminal control area.

In the tested ASP instances, the landing aircraft use both runways, while the take-off aircraft only used one (interdependent) runway resource for each TCA (i.e. runway resource 16 in Fig. 1 and runway resource 12 in Fig. 2). We do not perform rerouting measures for landing and take-off aircraft. D'Ariano et al. [27] and Samà et al. [66] focus on the combined aircraft reordering and rerouting problem at busy TCAs.

4. Problem formulation

In the general job shop scheduling formulation of the ASP, an *operation* denotes the traversal of a resource (i.e. air segment, common glide path, runway, holding circle) by a job (i.e. aircraft). The sequence of operations related to an aircraft represents the (pre-defined) route associated with that aircraft. The variables of the ASP are the start time t_i of each operation i to be performed by an aircraft on a specific resource. A set of route timings is *conflict-free* if, for each pair of operations associated with the same resource, the minimum time separation constraints are satisfied.

The ASP is represented by the alternative graph model [55], since this approach permits an accurate and efficient representation of the ASP [26,27,65–67]. Let G = (N, F, A) be the graph composed of the following sets: $N = \{0, 1, ..., n\}$ is the set of *nodes*, where nodes 0 and *n* represent the start and the end operations of the schedule, while the other nodes are related to the start of the other operations; *F* is the set of *fixed directed arcs* that model the sequence of operations to be executed by an aircraft; *A* is the set of *alternative pairs* that model the sequencing decision and inter-aircraft safety separations. Each pair is composed of two alternative directed arcs.

Each node $i \in N$ of the graph is associated with the start time t_i of operation i, and corresponds to the entrance of the associated aircraft to the associated resource. By definition, the start time of the schedule is a known value, e.g. $t_0 = 0$, and the end time of the schedule is a variable t_n .

Each fixed directed arc $(i,j) \in F$ has a length w_{ij}^F , which is uniquely determined by *i* and *j*. The fixed arc length w_{ij}^F models a minimum processing time between the start of *i* and the start of *j*, such that $t_j \ge t_i + w_{ij}^F$. In particular, $\sigma(i)$ denotes the operation following *i* on its route. It follows that $(i, \sigma(i)) \in F$ is the directed fixed arc connecting *i* with $\sigma(i)$ and $t_{\sigma(i)} \ge t_i + w_{i\sigma(i)}^F$.

Each alternative pair $((i, j), (h, k)) \in A$ has two arcs with length w_{ij}^A and w_{hk}^A . The alternative arc length w_{ij}^A represents a minimum separation time between the start of *i* and the start *j*. In particular, w_{ij}^A (w_{hk}^A) can be sequence-dependent, when nodes *i* and *j* (*h* and *k*) are operations of different jobs. Also, there can be multiple alternative arcs between nodes *i* and *j*.



Fig. 3. Alternative graph of a holding circle (a), selected (b) and not selected (c).

A selection *S* is a set of alternative arcs, at most one from each pair. An ASP solution is a *complete* selection S^c , where an arc for each alternative pair of the set *A* is selected, in which the connected graph (N, F, S^c) has no positive length cycles. Note that a positive length cycle represents an operation preceding itself, which is an *infeasibility*. Given a feasible schedule S^c , a timing t_i for operation *i* is the length of a longest path from 0 to *i* ($l^{S^c}(0, i)$). When minimizing a makespan-like objective function, an arc (k,n) between the end node *k* of each job and node *n* is added to the alternative graph, and a selection S^c is optimal if $l^{S^c}(0, n)$ is minimum over all the solutions.

The alternative graph can be viewed as a particular *disjunctive program*. We let *X* be the set:

$$X = \begin{cases} t_{\sigma(i)} - t_i \ge w_{\sigma(i)}^F & \forall (i, \sigma(i)) \in F \text{ with } \sigma(i) \neq n \\ t \ge 0, x \in \{0, 1\}^{|A|} : t_j - t_i + M(1 - x_{ij}) \ge w_{ij}^A \\ t_k - t_h + Mx_{ij} \ge w_{hk}^A & \forall ((i, j), (h, k)) \in A \end{cases}$$
(1)

The variables of the ASP are the following: |N| real variables t_i associated with the start time of each operation $i \in N$ and |A| binary variables x_{ij} associated with each alternative pair $((i, j), (h, k)) \in A$. The variable x_{ij} is 1 if $(i, j) \in S$, and $x_{ij} = 0$ if $(h, k) \in S$. The constant M is a sufficiently large number, e.g. the sum of all arc lengths.

The next subsections describe how the different types of TCA resources are modelled via alternative graphs, and show how each specific objective function can be formulated. A numerical ASP example of a traffic situation is illustrated for the FCO airport. For the proposed example, we give the trade-off between a set of non-dominated solutions, each one computed by solving a specific objective function to optimality. Graphs of the example are reported in Appendices A and B.

4.1. Resources in the alternative graph model

The TCA is composed of various types of resources. This section illustrates how each of them is modelled in the alternative graph. Fixed arcs are depicted with solid arrows and alternative arcs are depicted with dotted arrows.

Fig. 3(a) illustrates the formulation of a holding circle resource. We recall that holding circles are used by traffic controllers to let arriving aircraft wait before the start of their landing procedure when the TCA is congested. Let *i* be the entrance of aircraft A to the holding circle and let $\sigma(i)$ be the start of the next operation, the holding circle resource is formulated by two fixed arcs $(i, \sigma(i))$ and $(\sigma(i), i)$ (the two solid arrows), respectively of length $w_{i\sigma(i)}^F = 0$ and $w_{\sigma(i)i}^F = -\delta$, where δ is the time required to perform a holding circle, plus a pair of alternative arcs $((i, \sigma(i)), (\sigma(i), i))$ (the two dotted arrows) respectively of length $w_{i\sigma(i)}^A = \delta$ and $w_{\sigma(i)i}^A = 0$. The formulation of multiple (half) circles can be easily done in a similar way.

Fig. 3 (b) shows the decision to perform a holding circle (the dotted arrow $(i, \sigma(i))$ is selected), while Fig. 3(c) the case with no holding circle (the dotted arrow $(\sigma(i), i)$ is selected). The formulation of the holding circle constraints and variables is as follows:

$$t_{\sigma(i)} - t_i \ge w_{i\sigma(i)}^F = 0$$

$$t_i - t_{\sigma(i)} \ge w_{\sigma(i)i}^F = -\delta$$

$$t_{\sigma(i)} - t_i + M(1 - x_{i\sigma(i)}) \ge w_{i\sigma(i)}^A = \delta$$

$$t_i - t_{\sigma(i)} + M x_{i\sigma(i)} \ge w_{\sigma(i)i}^A = 0$$
(2)

If $x_{i\sigma(i)} = 0$ then no holding circle is performed, as shown in Fig. 3(c). Fig. 4(a) presents the formulation of an air segment. We define operation $i(\sigma(i))$ as the entrance (exit) of aircraft A to the air segment. The processing time of aircraft A in the air segment must be included in a range $[\theta_{min}, \theta_{max}]$. To model this range of values two fixed arcs $(i, \sigma(i))$ and $(\sigma(i), i)$ are used with length $w_{i\sigma(i)}^F = \theta_{min}$ and $w_{\sigma(i)i}^F = -\theta_{max}$, respectively. The following constraints are thus required for the processing time of aircraft A and B:

$$t_{\sigma(i)} - t_{i} \ge w_{i\sigma(i)}^{F} = \theta_{\min}^{A}$$

$$t_{i} - t_{\sigma(i)} \ge w_{\sigma(i)i}^{F} = -\theta_{\max}^{A}$$

$$t_{\sigma(j)} - t_{j} \ge w_{j\sigma(j)}^{F} = \theta_{\min}^{B}$$

$$t_{j} - t_{\sigma(j)} \ge w_{\sigma(j)j}^{F} = -\theta_{\max}^{B}$$
(3)

In each landing/take-off air segment, air traffic regulations impose a minimal longitudinal and diagonal separation distance between consecutive aircraft, that varies according to the aircraft category. The minimum separation time between two aircraft (named A and B in Fig. 4) in the same air segment is thus formulated as a sequence-dependent setup time. Since an overtake between two aircraft in the same air segment is not allowed for safety reasons, the entrance and exit orders between aircraft A and B must be the same. A sequencedependent setup time between aircraft A and B is required at the entrance and exit of the air segment as follows: if A precedes B (B precedes A) the setup time at the entrance of the air segment is $\mu_{in^{AB}}(\mu_{int^{BA}})$ and the setup time at the exit of the air segment is $\mu_{out^{AB}}(\mu_{out^{BA}})$. The sequencing decision variables are the two pairs of alternative arcs in Fig. 4(a): $((i, j), (\sigma(j), \sigma(i))$ with lengths $w_{ij}^{A} = \mu_{int^{AB}}$ and $w_{\sigma(j)\sigma(i)}^{A} = \mu_{out^{BA}}$.

In a feasible schedule (i.e. a complete selection with no positive length cycles), the selection of the two alternative arcs (i,j) and $(\sigma(i), \sigma(j))$ sequences aircraft A before B, i.e. the two constraints t_j $-t_i \ge w_{ij}^A$ and $t_{\sigma(j)} - t_{\sigma(i)} \ge w_{\sigma(i)\sigma(j)}^A$ are inserted in the graph. Otherwise, the two alternative arcs (j,i) and $(\sigma(j), \sigma(i))$ must be selected (aircraft B before A).

The formulation of the air segment sequencing decisions is next shown:

$$t_{j} - t_{i} + M(1 - x_{ij}) \ge W_{ij}^{A} = \mu_{in^{AB}}$$

$$t_{\sigma(i)} - t_{\sigma(j)} + Mx_{ij} \ge W_{\sigma(j)\sigma(i)}^{A} = \mu_{out^{BA}}$$

$$t_{\sigma(j)} - t_{\sigma(i)} + M(1 - x_{\sigma(i)\sigma(j)}) \ge W_{\sigma(i)\sigma(j)}^{A} = \mu_{out^{AB}}$$

$$t_{i} - t_{j} + Mx_{\sigma(i)\sigma(j)} \ge W_{ji}^{A} = \mu_{in^{BA}}$$
(4)

If $x_{ij} = 0$ then the alternative arc of length $w^A_{\sigma(j)\sigma(i)}$ is selected, i.e. aircraft B is scheduled first in the air segment (as shown in Fig. 4 (b)). Consequently, the variable $x_{\sigma(i)\sigma(j)}$ must be set to 0 (i.e. the alternative arc of length w^A_{ji} is selected); otherwise the alternative arc of length $w^A_{\sigma(i)\sigma(j)}$ is selected and a positive length cycle between nodes $\sigma(i)$ and $\sigma(j)$ is generated in the graph. In fact, there are only two possible sequencing decisions (A–B or B–A) between the two aircraft in the air segment, and therefore the variable $x_{\sigma(i)\sigma(j)}$ could be replaced with x_{ij} in (4).

Fig. 5(a) shows the runway formulation. Let *i*, *j* be the entrance of aircraft A, B to the runway and $\sigma(i)$, $\sigma(j)$ their exit. The processing times of aircraft A and B in the runway are π_A and π_B . These processing times are modelled by the two fixed arcs $(i, \sigma(i))$ and $(j, \sigma(j))$ with lengths $w_{i\sigma(i)}^F = \pi_A$ and $w_{i\sigma(i)}^F = \pi_B$:

$$t_{\sigma(i)} - t_i \ge W_{i\sigma(i)}^F = \pi_A$$

$$t_{\sigma(j)} - t_j \ge W_{i\sigma(i)}^F = \pi_B$$
(5)

In a feasible schedule, a sequence-dependent setup time is required at the runway. If A precedes B (B precedes A) the setup time at the runway is η_{AB} (η_{BA}). Furthermore, the runway is a nostore resource, since only one aircraft at a time can be scheduled.



Fig. 4. Alternative graph of an air segment (a), and a feasible arc selection (b).

M. Samà et al. / Omega ■ (■■■) ■■=–■■■



Fig. 5. Alternative graph of a runway (a), and a possible arc selection (b).



Fig. 6. Release and due date arcs for a landing aircraft A.

The no-store formulation is:

$$t_{j} - t_{\sigma(i)} + M(1 - x_{\sigma(i)j}) \ge w_{\sigma(i)j}^{A} = \eta_{AB}$$

$$t_{i} - t_{\sigma(j)} + Mx_{\sigma(i)j} \ge w_{\sigma(j)i}^{A} = \eta_{BA}$$
(6)

The sequencing decision is modelled via the alternative pair $((\sigma(i),j), (\sigma(j),i))$, in which the alternative arc $(\sigma(i),j)$ has length $w_{\sigma(i)j}^A = \eta_{AB}$ and the alternative arc $(\sigma(j), i)$ has length $w_{\sigma(j)i}^A = \eta_{BA}$. In Fig. 5(b), aircraft A is scheduled before aircraft B and the alternative arc $(\sigma(i),j)$ is selected. In this case the binary variable $x_{\sigma(i)j} = 1$.

The delay of each aircraft can be measured by means of suitable due date times. In this paper, we consider two due date times for each landing aircraft, measured at the entrance of the TCA and at the entrance of the runway. Take-off aircraft have a single due date time at the runway. Both landing and take-off aircraft have a release time at the entrance of the TCA.

Fig. 6 presents the alternative graph modeling of the release and due date times for a landing aircraft A. Node *i* models the first operation of aircraft A in the TCA, while node *j* models its runway operation. The runway due date time d_j is the scheduled entrance time of aircraft A in the runway, modelled by the *runway due date arc* (*j*,*n*) between nodes *j* and *n* of length $w_{jn}^F = -d_j$. The entrance due date time d_i is the scheduled entrance time of aircraft A in the TCA, modelled by the *entrance due date arc* (*i*,*n*) between nodes *i* and *n* of length $w_{in}^F = -d_i$. The entrance time of aircraft A in the TCA is constrained by a release time r_i , that is the minimum time at which this aircraft can enter the TCA, modelled by the *release arc* (0, *i*) between nodes 0 and *i* of length $w_{0i}^F = r_i$. The formulation of release and due date arcs for aircraft A is as follows:

$$t_n - t_j \ge W_{jn}^F = -d_j$$

$$t_n - t_i \ge W_{in}^F = -d_i$$

$$t_i - t_0 \ge W_{0i}^F = r_i$$
(7)

Additional scheduling requirements can be modelled in the alternative graph when defining the weight of runway due date arcs. For instance, one can take into account integrated recovery strategies between the runway and ground scheduling solutions as follows. The runway due date arc (j,n) can have length $w_{jn}^F = -d_j - s_j$, in which s_j is a slack time that can have a positive (negative) value if one want to have a more (less) flexible due date time at the runway. Alternatively, one can use the *deadline arc* (j, 0) between nodes j and 0 of length $w_{j0}^F = -D_j$ in order to fix a

deadline time D_j to start processing the runway: $t_0 - t_j \ge -D_j$. For instance, the latter type of arc can be used to model a rigid coordination time requirement between the runway and ground scheduling solutions.

4.2. Formulations with different objective functions

This section presents the MILP formulations of the ASP used in this paper. For each formulation, we consider all problem constraints introduced in the previous section and we describe how the specific objective functions have been modelled.

The first group of formulations is related to aircraft delay minimization. Here, we use the following types of due date arcs: the entrance due date arcs are associated with the first operation of landing aircraft and measure their entrance delay in the TCA; the runway due date arcs are associated with the operation modelling the entrance/exit to/from a runway resource of landing/ take-off aircraft and measure the delay caused by the resolution of potential aircraft conflict in the TCA. Both types of due date arcs measure the delay generated by the scheduling decisions, i.e. by the selection of one alternative arc from each alternative pair of the graph.

The length of entrance and runway due date arcs are defined in the alternative graph as follows. We let *i* be the first operation of an aircraft in the TCA, γ_i be its scheduled entrance time and q_i be the entrance delay. We assume that the latter information is an input data for the traffic controller. For the landing aircraft, the entrance due date arc has length $-d_i = -(\gamma_i + q_i)$, and the consecutive delay at the entrance of the TCA is max{0, $t_i - d_i$ }.

The total exit delay is the sum of the unavoidable delay (which cannot be recovered by aircraft rescheduling, even if the aircraft travel in the TCA with their minimum processing time) plus the consecutive delay (which is required to solve potential conflicts).

We let *j* be the arriving/departing operation at/from a runway *r* of a landing/take-off aircraft A, δ_j be its scheduled arrival/departure time and τ_j be the earliest possible entrance/exit time to/from the runway *r*. For each node *j* of a landing aircraft, τ_j is computed as the sum of the release time plus the minimum processing time into the landing air segments. In case of a departing aircraft, τ_j is the sum of the release time plus the scheduled processing time in the runway. The total exit delay of aircraft A at *r* is $t_j - \delta_j$. Since we want to minimize the consecutive delay at *r*, the runway due date arc has length $-d_j = -\max{\tau_j, \delta_j}$. The unavoidable delay at the runway is max $\{0, \tau_j - \delta_j\}$, while the consecutive delay at the runway is max $\{0, t_j - d_j\}$.

The **MAX TARDINESS** is the formulation that minimizes the maximum consecutive delay both for the entrance and runway due dates, that is the largest deviation from the entrance and due date times due to the resolution of potential conflicts in the TCA during the time horizon considered. This objective function can be represented as the minimization on the longest path from node 0 to node n in the alternative graph with entrance and runway due date arcs [25,26]. We observe that all aircraft have the same relevance with this objective function, meaning that this is the most equitable approach. The formulation is next shown:

 $\min_{\substack{k \in \mathcal{K} \\ t_n - t_k \ge -d_k \quad \forall (k,n) \in F \\ \{x,t\} \in X \end{cases}}$ (8)

The **AVG TARDINESS** is the minimization of the average consecutive delay both for the entrance and runway due dates. The

M. Samà et al. / Omega ■ (■■■) ■■■–■■■



Fig. 7. Example situation with landing and take-off aircraft at FCO TCA.

AVG TARDINESS problem is formulated as follows:

 $\min \frac{1}{|K|} \sum_{k=1}^{|K|} z_k$ s.t. (9) $\forall (k, n) \in F$ $z_k - t_k \ge -d_k$ $\forall k \in K$ $z_k \ge 0$ $\{x, t\} \in X$

where |K| is the number of due date arcs in the alternative graph and z_k is a real variable associated with the due date $k \in K$.

The two objective functions reported above support two different aspects of delay minimization: the maximum tardiness is the most equitable approach (since it minimizes the largest consecutive delay), while the average tardiness has a more global approach (since its value takes into account the delay of all aircraft).

The following objective function takes into account the request by some airline companies to limit the number of delayed aircraft above given tolerance thresholds. This corresponds to minimizing the number of deadline violations that often translate into penalty costs, according to service contracts between the airline companies and the traffic control authorities. In our approach, in order to maximize the satisfaction of airline company requirements, all violations have the same weight and we minimize the number of aircraft that have a consecutive delay above a given threshold value $P \ge 0$ at the runways. A violation is thus measured when an aircraft enters the runway with a consecutive delay greater than P. We call this formulation TARDY JOBS P:

$$\min \sum_{f=1}^{U} v_{f}$$
s.t.
$$Mv_{f} - t_{f} \ge -d_{f} - P \quad \forall (f, n) \in F$$

$$v \in \{0, 1\}^{U}$$

$$\{x, t\} \in X$$

$$(10)$$

where U is the number of runway operations, f is a runway operation connected with the runway due date d_{f} , and v_{f} is a boolean variable indicating if a violation happens (1) or not (0).

The next two formulations take into account priorities between different types of aircraft. The first formulation is the classical weighted average, named here PRIORITY TARDINESS:

$$\min \frac{1}{|K|} \sum_{k=1}^{|K|} f_k z_k$$
s.t.
$$z_k - t_k \ge -d_k \qquad \forall (k,n) \in F$$

$$z_k \ge 0 \qquad \forall k \in K$$

$$\{x,t\} \in X$$

$$(11)$$

where f_k is the weight associated with the due date arc starting from the node k. In this work, the weights are associated with the priority classes described in Section 3.

The second formulation takes into account aircraft priorities and focuses on maximizing the equity between the aircraft of each priority class. Here, the equity is defined as the difference between the largest (D_{max}^{c}) and the smallest (D_{min}^{c}) consecutive delays between all aircraft of the same class $c \in C$. This formulation is named **PRIORITY EQUITY**:

$$\min \frac{1}{|C|} \sum_{c=1}^{|C|} (D_{max}^{c} - D_{min}^{c})$$
s.t.

$$z_{k} - t_{k} \ge -d_{k} \qquad \forall (k, n) \in F$$

$$z_{k} - D_{max}^{c} \le 0 \qquad \forall (k, n) \in F \land k \in J_{c} \quad \forall C$$

$$z_{k} - D_{min}^{c} \ge 0 \qquad \forall (k, n) \in F \land k \in J_{c} \quad \forall C$$

$$D_{min}^{c} \ge 0 \qquad \forall C$$

$$D_{max}^{c} \ge 0 \qquad \forall C$$

$$\{x, t\} \in X$$

$$(12)$$

where |C| is the number of classes and J_c is the set of aircraft belonging to class $c \in C$.

The last group of formulations is related to the maximization of throughput. To deal with throughput, we fix $d_k = 0 \forall k \in K$. The formulation MAX COMPLETION minimizes the exit time of the last aircraft from the runway. This formulation corresponds to the following makespan minimization problem:

min
$$t_n$$

s.t.

$$t_n - t_k \ge 0 \quad \forall (k, n) \in F$$

 $\{x, t\} \in X$
(13)

The AVG COMPLETION formulation minimizes the average exit time from the runway:

$$\min \frac{1}{|K|} \sum_{k=1}^{|K|} z_k$$

$$z_k - t_k \ge 0 \qquad \forall (k, n) \in F$$

$$z_k \ge 0 \qquad \forall k \in K$$

$$\{x, t\} \in X$$
(14)

4.3. A numerical example

This section describes a simple traffic situation at the Roma Fiumicino (FCO) TCA, highlighting the difference between the optimal solutions computed for each model of the previous section. For each solution, we provide the value of all the other performance indicators.

M. Samà et al. / Omega ■ (■■■) ■■==■■

Table 1

Optimal ASP solutions computed for each formulation of the numerical example.

Objective function	Tardiness		Completion	Completion		Tardy Jobs		
	Max	Avg	Priority	Max	Avg	P=0	P=300	Equity
Max Tardiness	223	93.2	913.3	1728	1601	3	0	111.5
Avg Tardiness	225	68.8	701.7	1725	1590.5	2	0	103.3
Priority Tardiness	364	85.7	310.7	1822	1613	2	1	123.5
Max Completion	225	87.0	1065.0	1725	1590.5	2	0	103.3
Avg Completion	225	87.0	1027.5	1725	1590.5	2	0	103.3
Tardy Jobs $P=0$	256	83.8	454.3	1752	1615.3	2	0	125.8
Tardy Jobs P=300	256	83.8	497.0	1752	1615.3	2	0	125.8
Priority Equity	225	68.8	701.7	1725	1590.5	2	0	103.3

Fig. 7 presents a schematic view of the TCA and provides the route of each aircraft: A and C are landing aircraft, while B and D are take-off aircraft. A and D are delayed aircraft, i.e. they have an entrance delay that changes their release time. The entrance delay of A is 170 and the one of D is 489. Furthermore, all aircraft have to use the same runway resource (12), since the other one (13) is not available. The presence of disturbances causes potential conflict in the TCA and, therefore, the ASP must be solved. The alternative graph of the traffic situation is shown in Appendix A.

Table 1 gives the optimal solution value for each objective function (one per row in bold), and the corresponding value for all other performance indicators. The optimal solution is obtained by solving the ASP formulations of Section 4.2. Regarding the formulations with aircraft priorities, we consider a different class for each aircraft as described in Section 3. According to the aircraft types and the delayed aircraft, the adopted weights are: $f_A=20$, $f_C=10$, $f_D=2$ and $f_B=1$. The bigger the weight, the higher the priority class.

From the results of Table 1, we observe that, even for this simple traffic situation with four aircraft, seven different solutions are obtained by the eight ASP formulations. Only Avg Tardiness and Priority Equity share the same solution. The different ASP solutions present interesting gaps in terms of the various performance indicators. For instance, looking at the first row of Table 1, Max Tardiness gives the lowest maximum consecutive delay at the cost of large average consecutive delays and number of tardy jobs P=0. In fact, the optimal solution obtained for Max Tardiness minimizes the longest path in the graph, that includes operations of aircraft B and D only. This also explains the high value obtained for Priority Tardiness and Priority Equity (B and D are the two aircraft with lower priorities). Avg Tardiness (second row of Table 1) gives better results than Max Tardiness in terms of average consecutive delays, number of tardy jobs P=0 and Priority Equity, while Priority Tardiness is still far from the optimal value (third row of Table 1). Priority Equity gives a good trade-off between Max Tardiness and Avg Tardiness, however its solution is a poor quality compared to Priority Tardiness. The optimal solution obtained for Priority Tardiness clearly schedules the aircraft on their order of priorities (A, C, D and B), causing a very large consecutive delay for aircraft B (see Columns 2 and 8). The values obtained for Max and Avg Completion (fifth and sixth columns of Table 1) are similar for most of the solutions. Tardy Jobs P=0outperforms Tardy Jobs P=300 and represents a compromise solution in terms of the other indicators. The optimal solution for Tardy Jobs P=0 is shown in Appendix B. Gantt charts of the different ASP solutions in Table 1 are also illustrated in Appendix B.

5. Experimental results

This section presents the computational results for the ASP formulations of Section 4.2. The tests have been performed in a laboratory environment. We consider real-world ASP instances for FCO and MXP. The ASP solutions are computed via the solver IBM ILOG CPLEX MIP 12.0, with a given time limit of computation. The experiments are executed on a processor Intel Dual Core E6550 (2.33 GHz), 2 GB of RAM and Windows XP.

5.1. Description of the ASP instances

The assumptions made in the data sets have been inspired by the current practice at the studied airports. In general, the proposed MILP formulations can deal with any routing combination at runways (i.e. all runways can be eventually modelled in mixedmode), and any prioritization of take-off and landing aircraft (i.e. different weights can be used in the objective functions).

For each terminal control area (MXP or FCO), we deal with practical-size instances of 30-min traffic optimization. All traffic optimizations start at time $t_0 = 0$. The tests have been performed in a laboratory environment by using real-world data from MXP and FCO TCAs. The studied ASP instances are characterized by two types of aircraft (named medium and heavy aircraft). Consequently, sequence-dependent setup times are modelled in each ASP instance. The processing and setup times are computed for each aircraft category according to standard descent and ascent profiles, disregarding the actual aircraft passenger and freight load. The release and due date times are computed from a reference timetable.

Only one runway is used in a mixed mode for each TCA (resource 16 at MXP and resource 12 at FCO). There are up to 13 (12) aircraft scheduled in each runway during half hour traffic optimization at MXP (FCO). This can be considered a dense traffic while comparing the traffic flows at MXP and FCO with the other Italian airports. The impact of studying a limited time horizon of traffic optimization is not investigated in this paper.

Table 2 describes the 20 ASP instances that we generated with random entrance delays (10 with negative exponential distribution and 10 with Gaussian distribution). The entrance delays are randomly generated for the first half aircraft entering the TCA according to a given distribution. The exit delays are measured as a positive deviation from the scheduled take-off/landing time, that is derived from the reference timetable. Each row reports average information on a terminal control area. In total, the computational study is based on 40 ASP instances.

Table 2 is organized as follows. Column 1 presents the TCA (FCO and MXP), Columns 2 and 3 the number of landing and take-off aircraft, Columns 4 and 5 the maximum and average entrance delays (in seconds), Columns 6 and 7 the maximum and average unavoidable delays (in seconds). The latter delays are significantly

Table 2 ASP instances.

TCA	Landing aircraft	Take-off aircraft	Max entr. delay (s)	Avg entr. delay (s)	Max unavoid. delay (s)	Avg unavoid. delay (s)
MXP	14	6	1792.9	869.6	1476.7	683.5
FCO	16	4	1789.0	990.4	1638.9	783.7

Table 3

Variables and constraints for each ASP formulation.

ASP instance size	Variables		Constraint	Constraints	
TCA	MXP	FCO	MXP	FCO	
Max Tardiness Avg Tardiness Priority Equity Priority Tardiness Max Completion Avg Completion Tardy Jobs P=0s	658 692 700 692 658 692 678	806 842 850 842 806 842 826	1307 1341 1409 1341 1307 1341 1327	1605 1641 1713 1641 1605 1641 1625	

Table 4

Solutions for the ASP formulations.

TCA	MXP	FCO	MXP	FCO
Obj. function	Max Tardiness		Avg Tardiness	
Comp. time (s)	455.4	197.3	3483.0	2749.3
No. of optimal sol.	20	20	16	20
UB-1 (s)	389.3	324.6	69.0	55.0
UB-2 (s)	389.3	324.6	69.0	54.6
LB (s)	389.3	324.6	62.3	54.6
Gap-1 (%)	0	0	10.7	0.8
Gap-2 (%)	0	0	10.7	0
Obj. function	Priority Equ	iity	Priority Tardi	iness
Comp. time (s)	11.4	77.5	329.5	1698.4
No. of optimal sol.	20	20	20	20
UB-1 (s)	129.4	106.9	878.4	658.4
UB-2(s)	129.4	106.9	878.0	652.1
LB (s)	129.4	106.9	878.0	652.1
Gap-1 (%)	0	0	0.1	0.9
Gap-2 (%)	0	0	0	0
Obi. function	Max Comple	etion	Avg Completi	on
Comp. time (s)	529.7	2143.8	4208.1	2369.0
No. of optimal sol.	20	20	15	20
UB-1(s)	3363.3	3485.4	2463.5	2670.9
IIB-2(s)	3363.3	3475.3	2463.5	2670.9
IB(s)	3363 3	3475 3	2452.1	2670.9
Gan-1 (%)	0	03	0.5	0
Gap-2 (%)	0	0	0.5	0
Sup 2 (10)				
Obj. function	Tardy Jobs I	$\mathbf{r} = 0\mathbf{s}$	Tardy Jobs P	= 300s
Comp. time (s)	1.3	18.4	37.6	243.6
No. of optimal sol.	20	20	20	20
UB-1 (s)	7.5	8.4	1.5	1.2
UB-2 (s)	7.5	8.4	1.5	1.2
LB (s)	7.5	8.4	1.5	1.2
Gap-1 (%)	0	0	0	0
Gap-2 (%)	0	0	0	0

smaller than the entrance delays, since we compute the free-net processing times by letting all landing aircraft travel with their maximum allowed speed profile.

Table 3 gives the number of variables and constraints for each ASP formulation. Each row reports average information for the 20 ASP instances we considered for each airport (MXP or FCO). The FCO instances have more variables and constraints than the MXP instances, since more aircraft are scheduled at FCO and less alternative air segment resources are available for landing aircraft.

11

The next subsections will show the computational results obtained for the 40 ASP instances. We tested the 8 ASP formulations: Max and Avg Tardiness dealing with pure delay minimization, Priority Tardiness and Priority Equity dealing with aircraft classes, Max and Avg Completion dealing with throughput minimization, Tardy Job P=0 and Tardy Job P=300 dealing with deadline violations. In addition, we tested the 40 ASP instances with a practical scheduling rule and a combined formulation. As described in Section 3, we used four classes of aircraft: (1) landing, delayed aircraft; (2) landing aircraft on time; (3) take-off, delayed aircraft; (4) take-off aircraft on time. Their weights are: $f_1 = 20$, $f_2 = 10$, $f_3 = 2$ and $f_4 = 1$.

5.2. ASP formulations

Table 4 presents the average computational results obtained for each ASP formulation. Each column reports the average results on the 20 ASP instances for a traffic control area. Table 4 is organized in blocks of eight rows per formulation: Row 1 gives the objective function, Row 2 the average computation time (in seconds), Row 3 the number of problems that were solved to optimality by CPLEX, Row 4 the average upper bound value (in seconds, named UB-1) at up to 60 s, Row 5 the average upper bound value (in seconds, named UB-2) at up to 10,800 s. The experiments with a large computation time allowed us to get further information on the optimal solutions. The best known value of upper bounds is reported in bold. Row 6 gives the average best known value of lower bound (in seconds, named LB) obtained with the largest computation time, Row 7 the average optimality gap (in percentage, named GAP-1) computed as follows: (UB-1 - LB)/LB, Row 8 the average optimality gap (in percentage, named GAP-2) computed as follows: (UB-2-LB)/LB.

From Table 4, we have the following observations. All ASP instances are solved to (near)optimality (the optimality gap is always below 1%, except for the average tardiness at MXP airport showing around 10% optimality gap). The results obtained at 60 s of computation are similar to the ones obtained with larger computation times (comparing UB-1 versus UB-2). We conclude that the ASP instances can be efficiently solved in a short computation time (up to 60 s on a standard processor), compatible with real-time application.

Regarding the specific performance of the various objective functions, the Tardy Jobs formulations present the larger number of optimal solutions, while Avg Tardiness and Avg Completion the lowest number. More optimal solutions are generally obtained for the objective functions based on a maximum delay minimization compared to the ones based on an average delay minimization. Also, Priority Equity presents a larger number of optimal solutions than Priority Tardiness. The problem of minimizing the maximum consecutive delay, even if this is done per priority class, is thus easier to solve to optimality by CPLEX than the weighted average consecutive delay minimization.

5.3. Optimizing an objective and looking at the other objectives

This subsection studies how the optimal ASP solutions computed for a specific indicator infer the quality of the other indicators. The proposed analysis permits to assess the quality of nondominated solutions for one objective function in terms of the other performance indicators.

Fig. 8 presents average results on the 35/40 ASP instances solved to optimality by all ASP formulations. Each plot in this figure reports the average optimality gap in terms of all performance indicators that are not directly optimized in the objective function of the corresponding ASP formulation. The average optimality gap is computed for each secondary indicator as

M. Samà et al. / Omega ■ (■■■) ■■■–■■■



Fig. 8. Optimal solutions for an indicator viewed in terms of the other indicators.

follows: (value obtained for the secondary indicator – optimal value for that indicator)/optimal value for that indicator. For instance, the top-left plot reports the average results obtained for the Max Tardiness formulation, in which Avg Tardiness has 105% optimality gap, Priority Equity 41% and so on.

From Fig. 8, Avg Tardiness presents the best compromise solutions, since the optimality gap is, on average, up to 70% for all secondary indicators. In general, the objective functions based on a maximum delay minimization present larger gaps compared to the ones based on an average delay minimization (see left versus right plots). Specifically, Avg Tardiness presents a gap of 28% for Max Tardiness and smaller gaps than Max Tardiness for all secondary indicators but Priority Equity and Tardy Jobs P=0s. The

combination of Avg and Max Tardiness is thus worthy of investigation.

Regarding the objective functions based on classes and weights, optimizing with priorities does not cause a serious drop of the related performance indicators. In fact, Priority Equity presents good values of Max Tardiness, that is the most equitable performance indicator. Similarly, Priority Tardiness gives good values in terms of Avg Tardiness.

When looking at the throughput as a secondary indicator, most of the objective functions have performance very close to the optimal values of Max Completion and Avg Completion. We conclude that these indicators can easily be taken into account by the studied ASP formulations.

12

M. Samà et al. / Omega ■ (■■■) ■■■–■■■

Table 5			
Solutions for the ASP	when using	the FCFS	scheduling rule.

Objective	MXP TCA		FCO TCA		
Function	UB (sec)	N. of Optimal Sol.	UB (sec)	N. of Optimal Sol.	
Max Tardiness	529.4	4	493.6	3	
Avg Tardiness	128.2	0	115.1	0	
Priority Equity	212.5	0	148.8	1	
Priority Tardiness	2302.0	0	2230.2	0	
Max Completion	3458.1	7	3657.6	9	
Avg Completion	2990.4	0	2755.3	0	
Tardy Jobs P=0	13.2	0	13.4	0	
Tardy Jobs $P = 300$	49	6	53	10	



Tardy Jobs are among the less equitable objective functions, since they prefer to significantly penalize the behavior of a few aircraft in order to have the set of tardy aircraft as small as possible. Comparing the two corresponding formulations in terms of the other performance indicators, Tardy Jobs P=300s outperforms Tardy Jobs P=0s. In general, the latter formulation presents the worst values in terms of the various performance indicators.

It is interesting to note that, even if the considered objective functions are based on apparently similar performance indicators, the solutions optimized for a single indicator have the drawback to deteriorate the performance related to some of the other performance indicators.

5.4. A practical scheduling rule

This subsection studies the quality of the solutions computed by a commonly used scheduling rule: First Come First Served (FCFS). We observe that more elaborated approaches than FCFS, such as algorithms based on max position shifting [10,29,30], can be a more fairer comparison against real world behavior, since controllers usually modify the FCFS solution by switching groups of two or more aircraft around in order to improve throughput on some bottleneck resources.

Table 5 shows the average value (named UB, in seconds) on the 40 ASP instances of Section 5.1 obtained for the FCFS rule, and the number of optimal solutions for each performance indicator. The average computation time is less than 1 s.

Fig. 9 presents a plot with average results on 35/40 instances of Section 5.1 for which an optimal ASP solution has been computed for all performance indicators. The average optimality gap is computed for each performance indicator as in Section 5.3.

Overall, FCFS has a significantly larger optimality gap for all performance indicators compared to Avg Tardiness. This result highlights the importance to evaluate the ASP solutions in terms of several performance indicators, and to introduce new decision

Tabl	e 6	
------	-----	--

Solutions for the ASP when using the combined formulation.

Objective function	MXP		FCO	
	UB (s)	No. of optimal sol.	UB (s)	No. of optimal sol.
Max Tardiness	470.3	1	389.4	2
Avg Tardiness	73.1	6	57.2	5
Priority Equity	182.4	0	148.0	0
Priority Tardiness	1229.3	0	802.5	2
Max Completion	3377.4	7	3529.5	9
Avg Completion	2468.3	4	2676.0	4
Tardy Jobs $P=0$	11.8	0	12.5	0
Tardy Jobs P=300	3.3	5	1.7	14



Fig. 10. Combined formulation solutions in terms of all indicators.

support systems for solving the ASP based on the consideration and optimization of a number of performance indicators.

5.5. Combined formulation

When dealing with several performance indicators, we believe that a weighted objectives approach is not reasonable, since it would be very difficult to identify the right value of the weights. For this reason, we present the following combined formulation. First of all, we identify the ASP formulation of Section 4 that has the best trade-off performance. Then, we extend that formulation by introducing a set of additional constraints related to the nonoptimized performance indicators. The additional constraints are generated via the following *constraint generation procedure*:

- Phase I : A constraint $\alpha_i \leq \beta_i$ is added regarding the value α_i of each non-optimized indicator *i* in order to limit the deterioration of its performance up to a given threshold value β_i .
- Phase II : The resulting ASP formulation is solved to (near) optimality, provided that any ASP solution exists for the given additional constraints.
- Phase III : If at least a feasible ASP schedule exists, the procedure returns the best solution computed by the solver (we use CPLEX with a time limit of 10,800 s). Otherwise, the new constraints introduced in phase I are revised by increasing the threshold value β_i related to the value α_i of each performance indicator *i*.

The procedure iterates the three phases till a feasible ASP schedule is found.

The reason for studying the combined formulation is to investigate whether ASP solutions exist with a limited deterioration of the non-optimized indicators. Furthermore, the combined formulation does not require to set parameters in the objective

function, while it would be difficult to identify the right value when using e.g. parameters in the objective function in order to weight the different performance indicators.

From the results in Section 5.3, Avg Tardiness presents the best trade-off performance, since it is the only ASP formulation that is, on average, up to 70% from the optimal solution of each performance indicator. We therefore applied the constraint generation procedure to this ASP formulation and each of the 40 ASP instances of Section 5.1. Initially, the threshold value β_i has been fixed equal to the best known value α_i^* of each secondary indicator i. At each iteration of the constraint generation procedure, the current threshold value β_i is increased of 0.1 α_i^* for each secondary indicator *i* until a feasible schedule is found. Specifically, the following number of feasible schedules is found: 7 for 30% increase of the initial threshold value, 10 for 40% increase, 13 for 50% increase, 3 for 60% increase, 4 for 70% increase, and 3 for 80% increase. The computation of a trade-off schedule is thus a complex problem and a significant relaxation to the best known value of some secondary indicator is required to compute a feasible schedule for the studied ASP instances.

Table 6 presents average results on the 40 ASP instances for the combined formulation generated by the constraint generation procedure, in terms of the same information reported in Table 5.

Fig. 10 presents the average optimality gap for each performance indicator as in Section 5.3. The plot presents the average results on 35/40 instances of Section 5.1 for which an optimal ASP solution has been computed for all performance indicators.

The computational results in Fig. 10 show that the best tradeoff solutions are now computed by the solver for the combined formulation. The average distance from the optimal solution is reduced in the three worst cases (55% for Tardy Jobs P=0s, 43% for Tardy Jobs P=300s and 39% for Priority Equity) compared to Avg Tardiness. However, this improvement is obtained at the cost of a light deterioration the performance of other performance indicators, including the one minimized in the objective function (that has the largest worsening, on average, up to 5%).

6. Conclusions and further research

This paper presents microscopic formulations of the ASP with high precision modeling and the evaluation of alternative objective functions. We examine the trade-off between some classical performance indicators in the take-off and landing aircraft scheduling literature, since we believe that there is still not a generally recognized objective function for the ASP. We observe that optimal solutions are very often computed by a commercial solver within one minute of computation on a standard processor. Furthermore, the solver computes different ASP solutions for the ASP formulations with different objective functions. Finally, a pool of (near) optimal ASP solutions is provided to the traffic controller along with quantitative information on numerous performance indicators. An extensive set of computational experiments shows the existence of relevant gaps between the ASP solutions computed focusing on the studied aspects of the ASP. A trade-off between the quality of various performance indicators is found for the ASP solutions computed via the different formulations. In particular, the Avg Tardiness results to be a good trade-off formulation, and its solutions outperform the FCFS solutions in terms of all performance indicators. However, better trade-off solutions exist if one introduces additional constraints in Avg Tardiness.

In general, the development of an ASP formulation taking into account multiple performance indicators is a challenging problem. The solutions computed for a combined formulation may improve some indicator, while they may have the drawback to deteriorate the performance related to some other indicator. However, we believe that this work moves the interest of researchers and practitioners in paying more attention to the various modeling aspects and performance indicators related to the ASP, and thus on the inherent multi-objective nature of this problem.

Further research will be concentrated on developing real-time efficient (eventually heuristic) scheduling algorithms for specific ASP formulations that would (1) reduce the optimality gap found by CPLEX, (2) reduce the time to compute the best solution, (3) solve large-size ASP instances to (near)optimality. Furthermore, we intend to further improve the quality gaps in terms of the best known values of each indicator. The latter result can be achieved by investigating ASP formulations with multiple objectives, or by introducing additional ASP constraints while optimizing a single indicator (as shown in this paper for the combined formulation).

Other promising research directions should focus on the coordination of the ASP solutions with related problems, such as the en-route, ground and gate scheduling problems [32,59]. Additional factors should be considered, such as evaluating the impact of a dynamic system setting, integrating the ASP solutions with the ground scheduling solutions and the en-route scheduling solutions, dealing with other objectives, constraints and variables (e.g. aircraft routing and speed control).

Acknowledgment

This work was partially supported by the Italian Ministry of Research, Grant number RBIP06BZW8, project FIRB "Advanced tracking system in intermodal freight transportation". We are grateful to the associate editor and reviewers for their constructive and elaborated comments.

Appendix A

The traffic situation of Section 4.3 is modelled by the alternative graph of Fig. 11. Each node of the alternative graph represents an operation, e.g. A12 is aircraft A entering runway 12. Black solid arrows represent fixed directed arcs, while coloured dashed

M. Samà et al. / Omega ■ (■■■) ■■■–■■■



Fig. 11. Alternative graph for the numerical example. (For interpretation of the references to color in this figure caption, the reader is referred to the web version of this paper.)

arrows represent alternative directed arcs. The length of each arc is depicted in the graph for the routes of Fig. 7.

In this example there are 16 alternative pairs. For the two landing aircraft A and C, we assume that the capacity in the holding circle is unbounded, so there are no potential conflicts between them at the entrance of the TCA. However, we use four alternative pairs (two between nodes A1 and A4 and two between nodes C1 and C4) in order to model the possibility to perform one or two circles in the airborne holding. The shortest circle takes 180 (see the red alternative pairs ((A1,A4), (A4,A1)) and ((C1,C4), (C4, C1))), while the longest 240 (see the green alternative pairs ((A1, A4), (A4,A1)) and ((C1,C4), (C4,C1))). Since aircraft A and C have the same route in the TCA, four alternative pairs model the air segments: the (blue) pair ((A4,C4), (C9,A7)) and the (brown) pair ((C4, A4), (A7,C9)) for air segment 4; the (blue) pair ((A11,C11), (C12, A12)) and the (brown) pair ((C11,A11), (A12,C12)) for air segment 11. Another alternative pair between A and C is required on the runway resource: the (orange) pair ((Aout,C12), (Cout, A12)).

The two take-off aircraft B and D have a potential conflict on the runway resource with the landing aircraft. We thus have to use another five alternative pairs for resource 12: ((Aout,B12), (B14, A12)) (depicted in red), ((Aout,D12), (D14,A12)) (depicted in green), ((B14,C12), (Cout,B12)) (depicted in violet), ((B14,D12), (D14,B12)) (depicted in blue turquoise), ((Cout,D12), (D14,C12)) (depicted in grey). Aircraft B and D also have a potential conflict on air segment 14, that is modelled by the blue pair ((B14,D14), (Dout, Bout)) and the brown pair ((D14,B14), (Bout,Dout)).

Appendix **B**

Given the example of Section 4.3, Fig. 12 shows the optimal solution for the formulation "Tardy Jobs P=0". The alternative arcs selected in the solution are shown with coloured dashed arrows. The corresponding (complete) selection of the alternative arcs is the following: (A4,C4) and (A7,C9) (aircraft A is scheduled first on air segment 4); (A11,C11) and (A12, C12) (aircraft A is scheduled first on air segment 11); (C1,C4) of length 180 and (C4,C1) of length -180 (aircraft C must perform circles in the holding of length 180); (A4,A1) of length -180 and (A4,A1) of length 0 (aircraft A does not perform circles in the holding); (D14,A12), (D14,B12), (D14,C12), (Aout,B12), (Aout,C12) and (B14,C12) (the runway sequence is D - A - B - C); (D14,B14) and (Dout,Bout) (aircraft D is scheduled first on air segment 14).



Fig. 12. A feasible schedule for the numerical example.

M. Samà et al. / Omega **■** (**■■■**) **■■■■■**



Fig. 13. Gantt charts for the optimal ASP solutions for the numerical example.

16

M. Samà et al. / Omega ■ (■■■) ■■■–■■■

Fig. 13 illustrates the Gantt chart of the 7 optimal ASP solutions for the numerical example (we note that the optimal solution for Avg Tardiness and Priority Equity is the same ASP schedule).

References

- Adler N, Liebert V, Yazhemsky E. Benchmarking airports from a managerial perspective. Omega 2013;41(2):442–58.
- [2] Aktürk MS, Atamtürk A, Gürel S. Aircraft rescheduling with cruise speed control. Operations Research 2014;62(4):829–45.
- [3] Allahverdi A, Ng CT, Cheng TCE, Kovalyov MY. A survey of scheduling problems with setup times or costs. European Journal of Operations Research 2008;187(3):985–1032.
- [4] Andreatta G, Capanna L, De Giovanni L, Monaci M, Righi L. Efficiency and robustness in integrated airport apron, a support platform for intelligent airport ground handling. Journal of Intelligent Transportation Systems 2014;18 (1):121–30.
- [5] Andreatta G, Dell'Olmo P, Lulli G. An aggregate stochastic programming model for air traffic flow management. European Journal of Operations Research 2011;215(3):697–704.
- [6] Artiouchine K, Baptiste P, Dürr C. Runway sequencing with holding patterns. European Journal of Operational Research 2008;189(3):1254–66.
- [7] Atkin JAD, Burke EK, Greenwood JS, Reeson D. Hybrid metaheuristics to aid runway scheduling at London Heathrow airport. Transportation Science 2007;41(1):90–106.
- [8] Atkin JAD, Burke EK, Greenwood JS, Reeson D. On-line decision support for take-off runway scheduling with uncertain taxi times at London Heathrow airport. Journal of Scheduling 2008;11(1):323–46.
- [9] Atkin JAD, Maere GD, Burke EK, Greenwood JS. Addressing the pushback time allocation problem at Heathrow airport. Transportation Science 2013; 47(4):584–602.
- [10] Balakrishnan H, Chandran B. Algorithms for scheduling runway operations under constrained position shifting. Operations Research 2010;58(6):1650–65.
- [11] Ball MO, Barnhart C, Nemhauser G, Odoni A. Air transportation: irregular operations and control. In: Laporte G, Barnhart C, editors, Handbooks in operations research and management science, vol. 14; 2007. p. 1-67.
- [12] Ball MO, Hoffman R, Mukherjee A. Ground delay program planning under uncertainty based on the ration-by-distance principle. Transportation Science 2010;44(1):1–14.
- [13] Barnhart C, Bertsimas D, Caramanis C, Fearing D. Equitable and efficient coordination in traffic flow management. Transportation Science 2012; 46(2):262–80.
- [14] Barnhart C, Fearing D, Odoni A, Vaze V. Demand and capacity management in air transportation. EURO Journal of Transportation and Logistics 2012;1(1– 2):135–55.
- [15] Beasley JE, Krishnamoorthy M, Sharaiha YM, Abramson D. Displacement problem and dynamically scheduling aircraft landings. Journal of Operational Research Society 2004;55:54–64.
- [16] Beasley JE, Sonander J, Havelock P. Scheduling aircraft landings at London Heathrow using a population heuristic. Journal of Operational Research Society 2001;52(1):483–93.
- [17] Bertsimas D, Lulli G, Odoni A. An integer optimization approach to large-scale air traffic flow management. Operations Research 2011;59(1):211–27.
 [18] Bennell JA, Mesgarpour M, Potts CN. Airport runway scheduling. 40R-
- [18] Bennell JA, Mesgarpour M, Potts CN. Airport runway scheduling. 40R-Quarterly Journal of Operations Research 2011;9(2):115–38.
- [19] Bianco L, Dell'Olmo P, Giordani S. Minimizing total completion time subject to release dates and sequence-dependent processing times. Annals of Operations Research 1999;86:393–415.
- [20] Bianco L, Dell'Olmo P, Giordani S. Scheduling models for air traffic control in terminal areas. Journal of Scheduling 2006;9(3):180–97.
- [21] Castelli L, Pesenti R, Ranieri A. The design of a market mechanism to allocate air traffic flow management slots. Transportation Research—Part C 2011; 19(5):931–43.
- [22] Churchill AM, Lovell DJ, Ball MO. Flight delay propagation impact on strategic air traffic flow management. Transportation Research Records 2010;2177: 105–113.
- [23] Clare G, Richards AG. Optimization of taxiway routing and runway scheduling. IEEE Transactions on Intelligent Transportation Systems 2011;12(4):1000–13.
- [24] Clausen J, Larsen A, Larsen J. Disruption management in the airline industryconcepts, models and methods. Computers and Operations Research 2010; 35(5):809–21.
- [25] D'Ariano A, Pacciarelli D, Pranzo M. A branch and bound algorithm for scheduling trains in a railway network. European Journal of Operations Research 2007;183(2):643–57.
- [26] D'Ariano, A, D'Urgolo, P, Pacciarelli D, Pranzo M. Optimal sequencing of aircrafts take-off and landing at a busy airport. In: Proceedings of the 13th international ieee annual conference on intelligent transportation systems, Madeira Island, Portugal; 2010. p. 1569–74.
- [27] D'Ariano A, Pistelli M, Pacciarelli D. Aircraft retiming and rerouting in vicinity of airports. IET Intelligent Transport Systems 2012;6(4):433–43.
- [28] D'Ariano A, Pacciarelli D, Pistelli M, Pranzo M. Real-time scheduling of aircraft arrivals and departures in a terminal maneuvering area. Networks 2015; 65(3):212–27.

- [29] Dear, RG. The dynamic scheduling of aircraft in the near terminal area. Flight Transportation Laboratory (FTL) Report R76-9, Massachusetts Institute of Technology; 1976.
- [30] Dear RG, Sherif YS. An algorithm for computer assisted sequencing and scheduling of terminal area operations. Transportation Research—Part A 1991;25(2-3):129–39.
- [31] Diffenderfer P, Tao Z, Payton G. Automated integration of arrival/departure schedules. In: Proceedings of the 10th USA/Europe air traffic management research and development seminar (ATM 2013); 2013.
- [32] Dorndorf U, Drexl A, Nikulin Y, Pesch E. Flight gate scheduling: state-of-theart and recent developments. Omega 2007;35(3):326–34.
- [33] Duygun M, Prior D, Shaban M, Tortosa-Ausina E. Disentangling the European airlines efficiency puzzle: a network data envelopment analysis approach. Omega 2016;60(1):2-14.
- [34] Ernst AT, Krishnamoorthy M, Storer RH. Heuristic and exact algorithms for scheduling aircraft landings. Networks 1999;34(3):229–41.
- [35] Eun Y, Hwang I, Bang H. Optimal arrival flight sequencing and scheduling using discrete airborne delays. IEEE Transactions on Intelligent Transportation Systems 2010;11(2):359–73.
- [36] Eurocontrol. Airport collaborative decision-making. Accessed from (http:// www.euro-cdm.org/); October 2011.
- [37] Furini F, Kidd MP, Persiani CA, Toth P. Improved rolling horizon approaches to the aircraft sequencing problem. Journal of Scheduling 2015;18(5):435–47.
- [38] Glover CN, Ball MO. Stochastic optimization models for ground delay program planning with equity-efficiency trade-offs. Transportation Research—Part C 2013;33(1):196–202.
- [39] Gröflin H, Klinkert A. A new neighborhood and tabu search for the blocking job shop. Discrete Applied Mathematics 2009;157(17):3643–55.
- [40] Grushka-Cockayne Y, Reyck BD, Degraeve Z. An integrated decision-making approach for improving European air traffic management. Management Science 2008;54(8):1395–409.
- [41] Hall NG, Sriskandarajah C. A survey of machine scheduling problems with blocking and no-wait in process. Operations Research 1996;44(3):510–25.
- [42] Hasevoets N, Conroy P. Arrival Manager-Implementation guidelines and lessons learned. EUROCONTROL, Brussels. Technical report; 2010. p. 1–103.
- [43] Hu X-B, Chen W-H. Receding horizon control for aircraft arrival sequencing and scheduling. IEEE Transactions on Intelligent Transportation Systems 2005;6(2):189–97.
- [44] Hu X-B, Di Paolo E. An efficient genetic algorithm with uniform crossover for air traffic control. Computers and Operations Research 2009;36(1):245–59.
- [45] Kim J, Kröller A, Mitchell JSB, Sabhnani GR. Scheduling aircraft to reduce controller workload. In: Clausen J, Di Stefano G, editors, Proceedings of the 9th workshop on algorithmic approaches for transportation modeling, optimization, and systems (ATMOS 2009). Copenhagen, Denmark; 2009.
- [46] Kohl N, Larsen A, Larsen J, Ross A, Tiourine S. Airline disruption management perspectives, experiences and outlook. Journal of Air Transport Management 2007;13(3):149–62.
- [47] Kotynek B, Richetta O. Equitable models for the stochastic ground-holding problem under collaborative decision-making. Transportation Science 2006;40(1):133–46.
- [48] Kuchar JK, Yang LC. A review of conflict detection and resolution modeling methods. IEEE Transactions on Intelligent Transportation Systems 2000; 4(1):179–89.
- [49] Kuhn KD. Ground delay program planning: delay, equity, and computational complexity. Transportation Research Part C 2013;35(1):193–203.
- [50] Lamorgese L, Mannino C. An exact decomposition approach for the real-time train dispatching problem. Operations Research 2015;63(1):48–64.
- [51] Lieder A, Stolletz R. Scheduling aircraft take-offs and landings on interdependent and heterogeneous runways. Transportation Research Part E 2016;88:167–88.
- [52] Liu SQ, Kozan E. Scheduling trains with priorities: a no-wait blocking parallelmachine job-shop scheduling model. Transportation Science 2011;45(2): 175–198.
- [53] Manley B, Sherry L. Analysis of performance and equity in ground delay programs. Transportation Research—Part C 2010;18(6):910–20.
- [54] Marin AG. Airport management: taxi planning. Annals of Operations Research 2006;143(1):191–202.
- [55] Mascis A, Pacciarelli D. Job shop scheduling with blocking and no-wait constraints. European Journal of Operations Research 2002;143(3):498–517.
- [56] Mati Y, Xie X. Multiresource shop scheduling with resource flexibility and blocking. IEEE Transactions on Automation Science and Engineering 2011; 8(1):175–89.
- [57] Meloni C, Pacciarelli D, Pranzo M. A rollout metaheuristic for job shop scheduling problems. Annals of Operations Research 2004;131(1–4):215–35.
- [58] Mukherjee A, Hansen M. A dynamic stochastic model for the single airport ground holding problem. Transportation Science 2007;41(1):444–56.
- [59] Narciso ME, Piera MA. Robust gate assignment procedures from an airport management perspective. Omega 2015;50(1):82–95.
- [60] Pellegrini P, Castelli L, Pesenti R. Metaheuristic algorithms for the simultaneous slot allocation problem. IET Intelligent Transport Systems 2012;6(4):453–62.
- [61] Pellegrini P, Rodriguez J. Single European sky and single European railway area: a system level analysis of air and rail transportation. Transportation Research Part A 2013;57(1):64–86.
- [62] Pranzo M, Pacciarelli D. An iterated greedy metaheuristic for the blocking job shop scheduling problem. Journal of Heuristics Article 2016, <u>http://dx.doi.org/</u> <u>10.1007/s10732-014-9279-5</u> (in press).

M. Samà et al. / Omega ■ (■■■) ■■■–■■■

- [63] Psaraftis HN. A dynamic programming approach for sequencing identical groups of jobs. Operations Research 1980;28(6):1347–59.
- [64] Sabar NR, Kendall G. An iterated local search with multiple perturbation operators and time varying perturbation strength for the aircraft landing problem. Omega 2015;56(1):88–98.
- [65] Samà M, D'Ariano A, Pacciarelli D. Rolling horizon approach for aircraft scheduling in the terminal control area of busy airports. Transportation Research Part E 2013;60(1):140–55.
- [66] Samà M, D'Ariano A, D'Ariano P, Pacciarelli D. Optimal aircraft scheduling and routing at a terminal control area during disturbances. Transportation Research Part C 2014;47(1):61–85.
- [67] Samà M, D'Ariano A, D'Ariano P, Pacciarelli D. Air traffic optimization models for aircraft delay and travel time minimization in terminal control areas. Public Transport: Planning and Operations 2015;7(3):321–37.
- [68] Sherali HD, Hill JM, McCrea MV, Trani AA. Integrating slot exchange, safety, capacity, and equity mechanisms within an airspace flow program. Transportation Science 2011;45(2):271–84.
- [69] Sherali HD, Smith JC, Trani AA. An airspace planning model for selecting flightplans under workload, safety, and equity considerations. Transportation Science 2002;36(4):378–97.

- [70] Sherali HD, Staats RW, Trani AA. An airspace planning and collaborative decision-making model: Part i–probabilistic conflicts, workload, and equity considerations. Transportation Science 2003;37(4):434–56.
- [71] Sherali HD, Staats RW, Trani AA. An airspace-planning and collaborative decision-making model: part ii–cost model, data considerations, and computations. Transportation Science 2006;40(2):147–64.
- [72] Soomer MJ, Franx GJ. Scheduling aircraft landings using airlines' preferences. European Journal of Operations Research 2008;190(1):277–91.
- [73] Sölveling G, Solak S, Clarke JB, Johnson EL. Scheduling of runway operations for reduced environmental impact. Transportation Research Part D 2010; 16(2):110–20.
- [74] Tielrooij M, Borst C, Van Paassen M.M, Mulder M. Predicting arrival time uncertainty from actual flight information. In: Proceedings of the 11th USA/ Europe air traffic management research and development seminar (ATM 2015); 2015.
- [75] Venkatakrishnan CS, Barnett A, Odoni AM. Landings at Logan airport: describing and increasing airport capacity. Transportation Science 1993; 27(3):211–27.